Author's response to reviews

Title: The influence of a formula supplemented with dairy lipids and plant oils on the erythrocyte membrane omega-3 fatty acid profile in healthy full-term infants: a double-blind randomized controlled trial

Authors:

Maria L Gianni' (maria.gianni@unimi.it)
Paola Roggero (paola.roggero@unimi.it)
Charlotte Baudry (charlotte.baudry@lactalis.fr)
Amandine Ligneul (Amandine.ligneul@lactalis.fr)
Daniela Morniroli (daniela.morniroli@gmail.com)
Francesca Garbarino (francesca.garbarino@hotmail.it)
Pascale le Ruyet (pascale.leruyet@lactalis.fr)
Fabio Mosca (fabio.mosca@unimi.it)

Version: 2 Date: 22 August 2012

Author's response to reviews: see over
Cover Letter

Re: The influence of a formula supplemented with dairy lipids and plant oils on the erythrocyte membrane omega-3 fatty acid profile in healthy full-term infants: a double-blind randomized controlled trial.

Dear Editors,

Please find enclosed a revised version of the above referenced paper for your consideration in inclusion in *BMC Pediatrics* as a protocol article.

As requested, it has been cleared stated in the “competing interests” section of the manuscript that the study has received funding from a commercial organization (Lactalis Nutrition Santé, Torcé, France).

With regard to the ethical approval, we have uploaded the documents regarding the approval of the Ethical Committee of our Institution and their translation.

At the moment there are no manuscripts that are under submission with other journals based on this protocol.

We would like to thank the Editors for their helpful comments, and we hope that all remarks have been clarified.

We thank you for your consideration,

Maria Lorella Gianni, Paola Roggero, Charlotte Baudry, Amandine Ligneul, Daniela Morniroli,
Francesca Garbarino, Pascale le Ruyet, Fabio Mosca